CS360 Lab #1 Due 11:59PM, Tuesday, August 27, 1996

This assignment concentrates on using the fields.o set of procedures. See the lecture notes in the directory /blugreen/homes/plank/cs360/notes/Fields for instruction on how to use the

fields.o procedures.

mycat, myhead, mywc, and myfill

Implement mycat, myhead, mywc, and myfill, which are defined with the following man page. Use the fields.o set of procedures to read the input.

The first three of these programs are very simple (my versions are 53, 59, and 60 lines respectively). With mycat and myhead, you only need to access the line and text1 fields of the IS struct. With mywc, you only need to access the text1 and NF fields (and the line field if you want).

Myfill is more complex. Here, you’ll need to access the fields part of the IS struct, and do a

little more thinking. Still, the final program does not have to be too large (mine is 89 lines).

Error Conditions

Your program should flag an error if the user specifies an invalid command line argument, or a bogus one (like “myhead --9”, “myhead -jim”, “myfill -c 0” or “myfill -c -rj”). Error statements should be printed to stderr, and the program should exit with “exit(1)”.

Your program should do something reasonable if you encounter problemmatic input – such as a word with more than 72 characters. You should explain what your program does in such cases.

Working Programs

Working versions of mycat, myhead, mywc, and myfill are in the directory /blugreen/homes/plank/cs360/bin. If you have a question about what your program should do, first see what these programs do.

Useful Utilities

If you don’t know them already, you should read the man pages on strcpy, strlen, atoi, isdigit and sscanf. These can all be helpful to you.

1

2

3

4

5